Author’s response to reviews

Title: Trauma management training for community-based organizations in eastern Burma: Improving health worker skills and scaling up the workforce to address service delivery needs of communities in complex emergency settings. Allison Richard, MD1S, Catherine Lee, MPH 2, Matthew Richard, MD3, Eh Kalu Shwe Oo4, Thomas Lee, MD, MHS5, Lawrence Stock, MD6 1Assistant Professor of Emergency Medicine, Keck School of Medicine, Los Angeles, CA 2Field Director, Global Health Access Program. Mae Sot, Tak. Thailand 3Assistant Clinical Professor of Medicine, Geffen School of Medicine at UCLA, Los Angeles, CA 4Secretary, Karen Department of Health and Welfare, Mae Sot, Tak. Thailand 5Adjunct Associate Clinical Professor of Medicine, Geffen School of Medicine at UCLA, Los Angeles, CA 6Clinical Professor of Medicine, Geffen School of Medicine at UCLA, Los Angeles, CA SCorresponding author Email addresses: AR: sweetalmd@gmail.com CL: Catherine.i.lee@gmail.com MR: mattrichard@sbcglobal

Authors:

  Allison J Richard (sweetalmd@gmail.com)
  Catherine I Lee (catherine@ghap.org)
  Matthew G Richard (mattrichard@sbcglobal.net)
  Eh Kalu Shwe Oo (ehkalushweoo@gmail.com)
  Tom Lee (tomlee@ucla.edu)
  Larry Stock (laemdoc@aol.com)

Version: 2 Date: 1 October 2008

Author’s response to reviews:

Our reviewers provided several excellent suggestions. First, they recommended that the paper be refocused as a description of the history and development of the TMP, with a de-emphasis of the patient data collected by the TMP health workers. Changes were made to augment details of the TMP history, health worker background, and community context of the TMP.

Second, our reviewers suggested that the data collected on patient injuries and survival, while interesting, did not provide adequate basis to make any conclusions regarding the effect of the TMP on health care outcomes. This was due to the lack of baseline data as point for comparison. Thus, in the current draft, data is included, but purely as a descriptive feature.

Third, our reviewers felt that the TMP should not be compared to other trauma programs previously described in the literature, since these trauma programs differ too greatly in terms of health worker profiles, available resources, and community context. These comparisons were removed.